Author's response to reviews

**Title:** Transcript levels of Toll-like receptors 5, 8 and 9 correlate with inflammatory activity in Ulcerative Colitis

**Authors:**

Fausto Sanchez-Muñoz (fausto22@yahoo.com)
Gabriela Fonseca-Camarillo (gabrielafaster@gmail.com)
Marco A Villeda-Ramírez (nmavilledar_snk@hotmail.com)
Elizabeth Miranda-Perez (mirandapme@yahoo.com.mx)
Edgar J Mendivil (ed-jair@hotmail.com)
Rafael Barreto-Zuñiga (barretozu@yahoo.com)
Misael Uribe (muribe@rtn.net.mx)
Rafael Bojalil (bojraf@yahoo.com)
Aarón Domínguez-Lopez (aadominguezl@yahoo.com.mx)
Jesus K Yamamoto-Furusho (kazuofurusho@hotmail.com)

**Version:** 8  **Date:** 7 December 2011

**Author's response to reviews:** see over
Tim Shipley, PhD
BMC Gastroenterology
Editor in Chief

Dear Editor:

Enclosed you will find the final revised version of the manuscript: MS: 1380844289464145 - Transcript levels of Toll-like receptors 5, 8 and 9 correlate with inflammatory activity in Ulcerative Colitis. Author list has been corrected and we cropped all the figures so there is no white space around any of them.

Best regards,

Prof. Jesus K. Yamamoto-Furusho M.D., Ph.D.
Head of IBD Clinic
Department of Gastroenterology
Instituto Nacional de Ciencias Médicas y Nutrición